



This is an electronic reprint of the original article. This reprint may differ from the original in pagination and typographic detail.

Smidstrup, Søren; Stradi, Daniele; Wellendorff, Jess; Khomyakov, Petr A.; Vej-Hansen, Ulrik G.; Lee, Maeng Eun; Ghosh, Tushar; Jónsson, Elvar; Jónsson, Hannes; Stokbro, Kurt First-principles Green's-function method for surface calculations

Published in: Physical Review B

DOI: 10.1103/PhysRevB.96.195309

Published: 30/11/2017

Document Version Publisher's PDF, also known as Version of record

Please cite the original version: Smidstrup, S., Stradi, D., Wellendorff, J., Khomyakov, P. A., Vej-Hansen, U. G., Lee, M. E., Ghosh, T., Jónsson, E., Jónsson, H., & Stokbro, K. (2017). First-principles Green's-function method for surface calculations: A pseudopotential localized basis set approach. *Physical Review B*, *96*(19), 1-17. [195309]. https://doi.org/10.1103/PhysRevB.96.195309

This material is protected by copyright and other intellectual property rights, and duplication or sale of all or part of any of the repository collections is not permitted, except that material may be duplicated by you for your research use or educational purposes in electronic or print form. You must obtain permission for any other use. Electronic or print copies may not be offered, whether for sale or otherwise to anyone who is not an authorised user.

First-principles Green's-function method for surface calculations: A pseudopotential localized basis set approach

Søren Smidstrup,^{1,2} Daniele Stradi,^{1,*} Jess Wellendorff,¹ Petr A. Khomyakov,¹ Ulrik G. Vej-Hansen,¹ Maeng-Eun Lee,¹ Tushar Ghosh,³ Elvar Jónsson,^{2,3} Hannes Jónsson,^{2,3} and Kurt Stokbro¹

¹Synopsys QuantumWise, Fruebjergvej 3, DK-2100 Copenhagen, Denmark

²Faculty of Physical Sciences, University of Iceland VR-III, 107 Reykjavík, Iceland

³Department of Applied Physics, Aalto University, FI-00076 Espoo, Finland

(Received 11 July 2017; revised manuscript received 11 October 2017; published 30 November 2017)

We present an efficient implementation of a surface Green's-function method for atomistic modeling of surfaces within the framework of density functional theory using a pseudopotential localized basis set approach. In this method, the system is described as a truly semi-infinite solid with a surface region coupled to an electron reservoir, thereby overcoming several fundamental drawbacks of the traditional slab approach. The versatility of the method is demonstrated with several applications to surface physics and chemistry problems that are inherently difficult to address properly with the slab method, including metal work function calculations, band alignment in thin-film semiconductor heterostructures, surface states in metals and topological insulators, and surfaces in external electrical fields. Results obtained with the surface Green's-function method are compared to experimental measurements and slab calculations to demonstrate the accuracy of the approach.

DOI: 10.1103/PhysRevB.96.195309

I. INTRODUCTION

Atomic-scale modeling has established itself as a workhorse tool in computational materials science. Firstprinciples methods are routinely applied to study the physical and chemical properties of materials and material structures, including surface structures [1-6]. The slab approach to surface calculations, which models a surface structure with just a few atomic layers, has become the *de facto* standard for first-principles atomistic simulations of surfaces. This is despite the fact that a physical surface is a semi-infinite system, interfaced to the vacuum, unless the surface of an unsupported ultrathin film or membrane is considered.

A slab is by construction finite in the direction perpendicular to the surface plane, and it therefore has two surfaces, which are not always equivalent. As a consequence, the electronic structure of the surfaces of the slab is altered by quantum confinement along this out-of-plane direction. It means that the accuracy of the slab approach to modeling a semi-infinite surface may critically depend on the slab thickness [7,8]. This leads to a number of fundamental limitations on the applicability of the slab model for surface calculations. For example, converging surface properties such as work functions and surface energies with respect to the slab thickness is notoriously difficult [9,10], and using thin slabs can result in an inaccurate electronic structure for both metal [11], and semiconductor surfaces [12,13]. This drawback is well known with the cluster approach to modeling periodic systems, where the property of interest often exhibits a slow and sometimes cumbersome convergence behavior with respect to the cluster size [14,15].

Different alternative methods based on the surface Green'sfunction (SGF) formalism have been proposed to overcome the drawbacks of the slab approach to surface modeling [16–23]. In the SGF method, the semi-infinite system is divided into a finite surface region and a semi-infinite bulk region, as shown in Fig. 1. The bulk region acts as an electron reservoir, and the surface region is coupled to this bulk region through the selfenergy as discussed in Refs. [24–26]. The electronic structure of the entire surface system is calculated in a self-consistent manner, accounting for charge transfer between the bulk and surface regions, as well as for charge redistribution in the surface region. It means that the surface region becomes an open system interacting with the infinite reservoir of electrons that provides a physically correct description of a semi-infinite surface structure.

In spite of their advantages, the SGF-based methods have not found broad application in computational surface science, where the slab model continues to be the method of choice. This might be partly because one of the most popular implementations of density functional theory [27,28] (DFT) is based on the pseudopotential plane-wave basis set approach [29], which allows one to accurately converge the DFT calculations of material properties with respect to the basis set functions in a simple, systematic manner [30,31]. This approach is also computationally demanding for calculating large surface structures. In the linear combination of atomic orbitals (LCAO) approach, the Kohn-Sham (KS) single-particle Hamiltonian is represented in a tight-binding-like matrix form, which can be naturally adopted within the framework of the SGF formalism. The DFT calculations done with LCAO basis sets usually have a lower computational cost compared to that of the DFT plane-wave calculations since a relatively small number of localized basis functions is employed in practical calculations. That has its downside, as the use of too few basis functions may alter the computational accuracy.

Several implementations of the surface Green's-function formalism have been recently reported for both localized [22,32] and plane-wave basis set methods [33,34]. The latter takes advantage of a real-space representation for the Bloch states within the framework of the embedding method [35] or the maximally localized Wannier-function approach [36]. The computational issues discussed in the

^{*}stradi@synopsys.com



FIG. 1. Illustration of a typical semi-infinite surface configuration. The Hamiltonians of the surface region (\mathbf{H}_{S}) and the principal layer of the semi-infinite bulk region (\mathbf{H}_{B}), are coupled through the coupling terms \mathbf{V}_{SB} . A Dirichlet boundary condition is used at the boundary between the bulk and surface regions. In the vacuum region, a Neumann or Dirichlet boundary conditions can be used. L_{S} (L_{B}) is the length of the surface region (the bulk region principal layer) along the out-of-surface-plane (Z) direction.

previous paragraph still hold true for the plane-wave and LCAO-based SGF implementations, and need to be properly addressed to allow for both efficient and reliable SGF-based surface calculations.

In this paper, we present an efficient, accurate, selfconsistent SGF method for first-principles calculations of the total energy and electronic structure of surfaces that has been implemented in the ATOMISTIX TOOLKIT (ATK) simulation tool within the framework of the DFT pseudopotential LCAO basis set approach [37,38]. The present implementation of the SGF method takes an advantage of the highly optimized Green'sfunction methodology that has already been implemented in the ATK code for two-probe device simulations [24,39]. We develop new optimized LCAO basis sets (see Appendix) used in combination with recently developed SG15 optimized norm-conserving Vanderbilt pseudopotentials [40]. This allows for highly accurate LCAO calculations of material structure properties, with an accuracy similar to that of plane-wave based methods, and the computational efficiency of LCAO-based methods. This is of particular importance for an accurate description of the surface structures studied in our work.

We apply the ATK-SGF method to several surface problems that are inherently difficult to properly address with the traditional slab approach, including the calculation of metal work functions, band alignment in thin-film semiconductor heterostructures, surface states in metals and topological insulators, and the properties of adsorbates interacting with surfaces in external electric fields. For these studies, the ATK-SGF implementation has been combined with several methodological developments: (i) a real-space multigrid approach for imposing nonperiodic boundary conditions, e.g., for work function calculations or surface calculations with external electric field, (ii) an implementation of doping methods, e.g., for modeling doped semiconductor substrates [37], (iii) a pseudopotential projector-shift method for resolving the problems of DFT in describing correctly the band gap of semiconductors (see Appendix), (iv) an implementation of spin-orbit coupling, which is an important effect in topological insulators [41], and (v) self-consistent energy calculations directly within the SGF method, e.g., for studying adsorbates on metal surfaces.

The paper is organized as follows. Section II describes the methodology and basic computational settings adopted in this

work, as well as implementation details of the SGF method and its computational efficiency. Section III shows how to calculate work functions of metal surfaces that are well-converged with respect to the system size, using the SGF method. In Sec. IV, the SGF method is applied for understanding of the band alignment in a semiconductor heterostructure such as a Si film on doped Ge(001) substrates. Section V shows how the SGF method can be used to calculate pure surface states in metals and topological insulators. Section VI describes an application of the SGF method for surface chemistry problems such as the adsorption of iodine atoms on the Pt(111) surface in the presence of an external electric field. The main conclusions are summarized in Sec. VII.

II. METHODOLOGY

A. Electronic structure method

Our implementation of the surface Green's-function method is done within the framework of density functional theory [27,28,42,43] using the norm-conserving pseudopotential LCAO basis set approach [39,44]. The corresponding Kohn-Sham (KS) Hamiltonian can be written as

$$\hat{H}^{\rm KS} = -\frac{\hbar^2}{2m} \nabla^2 + V_{\rm loc} + V_{\rm nl} + V_{\rm H} + V_{\rm xc}, \qquad (1)$$

where the first term corresponds to the electron kinetic energy, V_{loc} and V_{nl} are the local and nonlocal parts of the pseudopotential, respectively, and the Hartree (V_{H}) and exchange-correlation (V_{xc}) potentials are given by the last two terms.

Using an LCAO basis allows representing the KS Hamiltonian in a matrix form with the following matrix elements [24,39]

$$\mathbf{H}_{ij}^{\mathrm{KS}} = \langle \phi_i(r) | \hat{H}^{\mathrm{KS}} | \phi_j(r) \rangle, \qquad (2)$$

where $\phi_i(r)$ and $\phi_j(r)$ are localized finite-range numerical orbitals [45,46]. To evaluate the Hamiltonian matrix elements in Eq. (2), we follow the SIESTA method [39], where the $V_{\rm H}$ and $V_{\rm xc}$ terms are calculated on a real-space grid.

B. Pseudopotentials and basis sets

Using a pseudopotential LCAO approach requires a careful choice of the pseudopotential and LCAO basis set to do

computationally efficient DFT calculations without compromising the accuracy of the obtained numerical results. We have developed three types of SG15 pseudopotential-based basis sets corresponding to Ultra, High, and Medium accuracy for all elements in the periodic table up to Z = 83. The SG15-Ultra basis sets provide the accuracy of DFT-LCAO calculations comparable to that of the state-of-the-art allelectron calculations, whereas the SG15-Medium basis set type allows for computationally cheap calculations with an error that is of the same order as that due to the use of approximate DFT functionals within the framework of local density (LDA) or generalized gradient approximations (GGA). Adopting the Medium basis set, we typically gain an order of magnitude in the computational efficiency compared to the Ultra basis set. In Appendix, we present the methodology to generate these basis sets, and benchmark the corresponding DFT-LCAO calculations against reference all-electron and pseudopotential plane-wave DFT calculations to evaluate the pseudopotential and basis set accuracy.

A reliable study of semiconductor physics problems usually requires an accurate description of the band gap. Unfortunately, the DFT approach based on local and semilocal DFT density functionals fails to accurately calculate the band gap of semiconductor materials [47]. To overcome this problem we have introduced a set of adjustable parameters for the pseudopotentials somewhat similar to the empirical pseudopotentials proposed by Zunger and co-workers [48]. This approach allows for a good description of both structural and electronic properties of semiconductors. This method has been used for studying a Si thin-film on the Ge(001) substrate in Sec. IV (more details on the generation of the parameters can be found in Appendix).

C. Green's-function method

Using a finite-range LCAO basis set allows for partitioning the Hamiltonian of the semi-infinite surface into three distinct matrix blocks that correspond to the Hamiltonian of the surface region (\mathbf{H}_{S}), a single atomic layer ("principal layer") of the semi-infinite bulk region (\mathbf{H}_{B}) and the coupling matrices (\mathbf{V}_{BB} and \mathbf{V}_{SB}), as illustrated in Fig. 1 [24]. The coupling matrices, \mathbf{V}_{SB} and \mathbf{V}_{BB} , account for interaction between the surface and bulk region atomic layers, and between the principal layers of the semi-infinite bulk region, respectively. In the ATK implementation, the coupling matrix, \mathbf{V}_{SB} , is expressed in terms of the \mathbf{V}_{BB} matrix as described in Ref. [24], assuming that a sufficiently thick layer of the material comprising the semi-infinite bulk region is added to the surface region. The infinite Hamiltonian matrix of the entire system can then be written as

$$\mathbf{H}^{\mathrm{KS}} = \begin{pmatrix} \ddots & \vdots & \vdots & \vdots & \vdots \\ \cdots & \mathbf{V}_{\mathrm{BB}}^{\dagger} & \mathbf{H}_{\mathrm{B}} & \mathbf{V}_{\mathrm{BB}} & \mathbf{0} \\ \cdots & \mathbf{0} & \mathbf{V}_{\mathrm{BB}}^{\dagger} & \mathbf{H}_{\mathrm{B}} & \mathbf{V}_{\mathrm{SB}} \\ \cdots & \mathbf{0} & \mathbf{0} & \mathbf{V}_{\mathrm{SB}}^{\dagger} & \mathbf{H}_{\mathrm{S}} \end{pmatrix}.$$
(3)

Using Green's-function formalism [49], the density matrix of the surface region, D_S , can be expressed as

$$\mathbf{D}_{\mathrm{S}} = -\frac{1}{\pi} \int_{-\infty}^{\mu_{\mathrm{B}}} \mathrm{Im}[\mathbf{G}_{\mathrm{S}}(\epsilon)] d\epsilon, \qquad (4)$$

where μ_B is the bulk chemical potential, and G_S is the finite Green's-function matrix of the surface region

$$\mathbf{G}_{\mathrm{S}}(\epsilon) = [(\epsilon + i\delta)\mathbf{S}_{\mathrm{S}} - \mathbf{H}_{\mathrm{S}} - \Sigma(\epsilon)]^{-1}, \qquad (5)$$

where S_S and H_S are the overlap and Hamiltonian matrices associated with the basis set functions centered inside the surface region, respectively; Σ is the self-energy matrix describing the coupling of the surface to the semi-infinite bulk region, i.e., accounting for open boundary conditions imposed on the surface region. In most cases, the initial guess for the Hamiltonian H_S can be constructed from a superposition of atomic densities. Obtaining the initial guess for H_S from a conventional calculation of a slab corresponding to the surface region is also possible for systems exhibiting difficult convergence behavior, which is the case of the calculation including noncollinear spin-orbit coupling carried out in this work for the Bi₂Se₃(111) surface, presented in Sec. V B.

Given the density matrix, the electron density, n(r), is constructed as

$$n(r) = \sum_{ij} [D_{\mathrm{S}}]_{ij} \phi_i(r) \phi_j(r) + n_{\mathrm{sp}}, \qquad (6)$$

where n_{sp} is the "spill-in" corrective term related to density matrix components in the bulk region and the bulk-surface boundary [25,37]. Including this term is crucial to describe correctly the charge density at the boundary between the surface and bulk regions, by accounting explicitly for the density in the surface region due to those basis functions in the bulk region, which tails penetrate into the surface region. For a more extensive description, we refer the reader to Ref. [37]. The electron density, the Hartree and exchange-correlation potentials and the surface Green's function can then be obtained by solving the Kohn-Sham and Poisson equations together with Eqs. (1)-(6) in a self-consistent manner, using a procedure equivalent to that described in Ref. [24], but for a system formed by a central region coupled to a single electron reservoir. Depending on the actual physical problem of study, the Poisson equation can be solved with the Dirichlet, or mixed (Dirichlet + Neumann) boundary conditions as shown in Fig. 1.

D. Implementation details

The numerical implementation of the SGF method is an extension of the development done for simulating two-terminal devices in the ATK [24,25]. In the SGF method, a single electron reservoir is only needed to impose the open boundary condition on the surface region. This means that the integral in Eq. (4) comprises only the equilibrium part of the Green's function, which can be efficiently evaluated using complex contour integration. Subsequently, the density matrix in Eq. (4) can be written as

$$\mathbf{D}_{\mathrm{S}} = \sum_{k} w_k \mathbf{G}_{\mathrm{S}}(z_k),\tag{7}$$

where the complex energies z_k and the weights w_k are determined as described elsewhere [24,25].

To calculate the Green's-function matrix, G_S , we have to compute the self-energy matrix (Σ) of the semi-infinite bulk region, which will be called the electrode in the following. For that, we first obtain the electrode matrices, H_B and V_{BB} ,

from a bulk calculation using periodic boundary conditions. The self-energy matrix in Eq. (5) can then be computed directly from propagating and evanescent modes [50,51], which can be efficiently calculated with an iterative method as proposed in Ref. [52]. Here we adopt a more efficient recursive method for the self-energy matrix calculation that does not require an explicit calculation of the electron modes in the bulk electrode [53]. Using the recursion method proposed in Ref. [53], we exploit the sparsity of the bulk Hamiltonian matrix, and find that this method gives the best balance between stability, accuracy, and computational efficiency.

The Green's-function matrix is eventually calculated with the Sweep method optimized for application to the surface configuration [54]. This method allows for finding the Green'sfunction matrix in O(N) steps, where N is the number of diagonal blocks in the block tridiagonal Hamiltonian matrix. The Hamiltonian matrix elements are preordered to give an optimal block tridiagonal structure [22]. Alternatively, the MUMPS [55] and PEXSI [56] libraries, which allow for lower memory consumption and parallel scaling to a larger number of computing processors, can also be employed. We find, however, that their serial performance is worse than that of the Sweep method, in general.

A significant advantage of using Green's-function techniques is that the complexity of the calculation scales as $O(M^{\alpha}N)$ instead of the typical $O(M^3N^3)$ scaling of DFT calculations using periodic boundary conditions, where 2 < $\alpha \leq 3$, and *M* is the dimension of the matrix corresponding to each of the N blocks in the block tridiagonal Hamiltonian matrix. The actual value of α depends on the particular implementation of matrix operations adopted for Green'sfunction matrix calculations. The time required for a single Green's-function SCF cycle therefore scales linearly with the number of surface atomic layers, instead of the usual cubic scaling. Figure 2 shows a comparison of the CPU time per self-consistent cycle needed to calculate a surface configuration of length $L_{\rm S}$ and a slab configuration having an equivalent length. For a length $L_{\rm S} = 13.5$ nm, corresponding approximately to the width of the depletion layer in bulk silicon



FIG. 2. CPU time per self-consistent (SCF) cycle as function of the length of the surface region L_S (see Fig. 1) for a surface configuration (red filled circles, solid line), compared to the CPU time per SCF cycle of a slab configuration (blue filled diamond, dashed line) with the slab thickness equivalent to L_S . The system considered is a 2 × 2 unreconstructed Si(100) surface.

at an *n*-doping level of $n = 10^{18}$ cm⁻³, one can see that a slab calculation is more computationally expensive than an SGF calculation by a factor of 5.

The Hartree potential term $V_{\rm H}$ in Eq. (1) is obtained by solving the Poisson equation with a Dirichlet boundary condition at the electrode-surface interface and a Neumann boundary condition in the vacuum. These mixed boundary conditions are exact for a semi-infinite surface in the absence of an external electric field. External fields can be included by imposing Dirichlet and Neumann boundary conditions also in the vacuum region, enabling simulations of surface structures in external electric fields. In both cases, the Poisson equation is solved using either a multigrid solver or the two-dimensional (2D) FFT method introduced in Ref. [25].

All time-demanding steps are parallelized in the ATK, including calculation of the Green's-function matrix in Eq. (5), the real-space density in Eq. (6), the real-space potentials in Eq. (1), and Hamiltonian in Eq. (2). In particular, the SGF calculations are parallelized over k points and contour integration points for the Green's-function matrix calculation.

E. Computational details

In this paper, the ATK-DFT calculations have been done using the GGA-PBE exchange-correlation functional [57] and the SG15-Medium combination of norm-conserving pseudopotentials and LCAO basis sets, unless otherwise stated. We have adopted a real-space grid density that is equivalent to a plane-wave kinetic energy cutoff of 100 Ha, and the Monkhorst-Pack k-point grids for the Brillouin zone sampling [58]. For the bulk electrodes, three-dimensional grids have been used to sample the 3D Brillouin zone. In order to properly converge the self-energy matrices Σ entering in Eq. (5), very dense k-point grids have been used in the direction normal to the surface plane [24]. For the SGF calculations, the system is periodic only along the directions parallel to the surface plane, so that 2D k-point grids have been used in this case. The choice of the actual k-point sampling depends on the system considered, and will be reported in each of the following sections. The broadening of the Fermi-Dirac distribution is chosen to be of 0.026 eV. The total energy and forces have been converged at least to $\sim 10^{-4}$ eV and $0.01 \text{ eV}/\text{\AA}$, respectively.

III. WORK FUNCTION CALCULATIONS

The work function W is a fundamental electronic property of a surface. Knowing the work function values for metal surfaces is of particular importance in electronics [61] and (photo)electrochemistry [62]. The work function is the energy required to remove an electron from the Fermi level (E_F) of a cleaved crystal to the vacuum level,

$$W = -e\phi - E_{\rm F},\tag{8}$$

where e is the elementary charge, e > 0, and ϕ is the electrostatic potential in the vacuum region near the surface plane.

Work function calculations based on the DFT approach most often employ a slab model for the surface structure. This often requires using a dipole correction to eliminate a spurious



FIG. 3. The macroscopic in-plane averaged electrostatic difference potential $\langle \delta V_{\rm E} \rangle$ [59] calculated throughout the surface and slab structures for the SGF model (blue solid line) and the ATK slab model (red solid line), respectively. For both surface models, the $\langle \delta V_{\rm E} \rangle$ potential is calculated for 14-monolayer Ag(001) slab or surface region, and is projected onto the Z axis (normal to the surface plane). Boundary conditions of Dirichlet or Neumann type are indicated for both SGF and slab models of the surface.

interaction between periodically repeated slab images [63]. Furthermore, the computed work function may converge slowly with respect to the number of atomic monolayers in the slab. So, accurate work function calculations can be computationally intensive within the framework of the slab approach. In this section, we demonstrate that employing the ATK-SGF based approach for calculating work functions of metal surfaces resolves these issues.

a. Methods. The fundamental difference between the slab and SGF methods for ATK-DFT surface calculations is illustrated in Fig. 3, which shows the macroscopic in-plane averaged electrostatic difference potential, $\langle \delta V_{\rm E} \rangle$ [59], throughout the Ag(001) surface structure for the slab model and the SGF model of the surface. Both model structures of the Ag(001) surface are effectively comprised of 14 atomic monolayers. There exists, however, a crucial difference between the slab and SGF-modeled surface structures, as the SGF-modeled Ag(001) surface region is matched to that of bulk Ag region as discussed in Sec. II.

For work function calculations, we impose a Dirichlet (Neumann) boundary condition on the right (left) side of the slab, see Fig. 3. It means that the electrostatic potential is zero near the surface in the vacuum on the right side of the slab, and the slab-calculated work function (W_{slab}) of the corresponding surface is given by the slab chemical potential $E_{\text{F}}^{\text{slab}}$,

$$W_{\rm slab} = -E_{\rm F}^{\rm slab}.$$
 (9)

For the SGF model, a Neumann (Dirichlet) boundary condition is adopted in the vacuum (at the interface between the surface and bulk regions) as shown in Fig. 3. In this case, the chemical potential of the entire surface system is that of the bulk region, and the SGF-calculated work function, W_{SGF} , is then given as

$$W_{\rm SGF} = -e \langle \delta V_{\rm E} \rangle^{\rm vac} - E_{\rm F}^{\rm bulk}, \qquad (10)$$

where $E_{\rm F}^{\rm bulk}$ is the Fermi energy of the bulk region and $\langle \delta V_{\rm E} \rangle^{\rm vac}$ is the macroscopic in-plane averaged electrostatic difference potential far away from the surface in the vacuum [64].



PHYSICAL REVIEW B 96, 195309 (2017)



FIG. 4. The Ag(001) work function calculated as a function of a number of atomic monolayers in the surface region, using the ATK-SGF method (red filled circles), the ATK (blue filled diamonds), and VASP (gray filled squares) slab models.

For the sake of comparison, we have calculated work functions using both the SGF and slab method. The slab model has been employed within the framework of the LCAO and plane-wave (PW) based approaches as implemented in the ATK and VASP codes, respectively [31,38]. We have used a 1×1 surface primitive cell and vacuum layers with a thickness of ~ 12 Å. The 2D Brillouin zone (BZ) of the surface has been sampled with a $15 \times 15 k$ -point grid, and a $15 \times 15 \times 101 k$ point grid has been adopted for sampling the 3D BZ of the bulk metal. We have done ion relaxation for the top layers of the metal surface, converging the forces to a maximum value of 0.01 eV/Å. For the ATK work function calculations, three ghost atoms have been added to the surface structure near the surface to accurately account for the electron density decaying into the vacuum [65]. All the other ATK computational details are given in Sec. IIE. For the VASP calculations, we have employed a kinetic energy cutoff of 400 eV and a dipole correction [63] in the out-of-surface-plane direction.

b. Results. Figure 4 shows how the Ag(001) work function, which is calculated using the slab (SGF) model, converges with respect to the number of atomic monolayers in the slab (surface region). This figure suggests that rather thick slabs are needed to converge the work function, whereas the SGF-calculated work function is almost independent of the surface region thickness. The main reason for this fast convergence is that the SGF-calculated electronic structure of the surface region is coupled to that of the semi-infinite bulk region, meaning that the bulk states are taken into account in an exact manner for any thickness of the surface region. In the slab approach, one would have to increase the slab thickness significantly to accurately describe the bulk states as seen in Fig. 4.

To demonstrate that the SGF method for work function calculations is accurate for various metal surfaces, we have computed the work functions of 11 transition metal surfaces such as the Ag(100), Au(111), Co(111), Cr(110), Cu(100), Ir(100), Pd(100), Pd(110), Pt(111), Rh(100), and Ru(001) surfaces. For the work function calculations, we have built metal slabs and surface regions with the thickness of 13 atomic monolayers, using experimental lattice parameters of bulk metals.

Figure 5 shows that the work functions calculated with the ATK-SGF and PW-slab approaches agree with the



FIG. 5. DFT-calculated vs experimental work functions. The work functions are calculated using the ATK-SGF method (red filled circles) and the VASP plane-wave slab method (gray filled squares). The measured work functions are taken from Ref. [60].

experimental data within a mean error of ~0.26 eV and an absolute error of ~0.5 eV [60]. This figure also suggests that the work function values calculated with the PW-slab approach are in a good agreement with the SGF-obtained work functions, provided sufficiently thick (13 atomic monolayers) slabs are adopted for the slab calculations. The absolute (mean) error between the SGF- and slab-calculated work functions is in the range of ~0.1 eV (~0.07 eV), which is smaller than the computational absolute (mean) error ~0.5 eV (~0.26 eV) estimated by comparing the calculated work functions to measured ones in Fig. 5.

In conclusion, we demonstrated that using the SGF method for work function calculations is more advantageous, compared to the slab method, as the SGF-calculated work function converges much faster with respect to the thickness of the surface model structure. The ATK-LCAO results obtained in this section suggested that the ATK-LCAO basis sets (see Appendix) combined with the SG15 optimized normconserving pseudopotentials [40] provide the accuracy of LCAO-based work function calculations that is similar to that of PW-based calculations.

IV. BAND ALIGNMENT IN SEMICONDUCTOR HETEROSTRUCTURES

In this section, we address the issue of how to calculate the electronic structure of a semiconductor surface in an accurate manner, and how the band alignment between the surface and a semiconducting thin-film can then be defined and calculated from first-principles atomistic simulations. We demonstrate that the SGF approach resolves several severe limitations of the slab approach for the band-structure calculations of semiconductor surfaces and interfaces.

A. Ge(001) surface

First, we study the Ge(001) surface, using the SGF approach and comparing it to the conventional slab approach. The bandstructure calculation of semiconductor surfaces is a challenging computational problem compared to that of metal surfaces since, among other effects, the semiconductor energy gap has a strong dependence on the slab thickness because of quantization effects. We show that the SGF approach allows one to overcome this particular drawback of the slab approach, accounting for the bulk semiconductor states in an exact manner by imposing open-boundary conditions on the semiconductor surface structure.

a. Methods. For the slab calculations of the Ge(001)surface, we have built a set of Ge(001) slabs with increasing thicknesses, L/a = 3, 4, 5, 6, and 7, where the lattice constant of bulk Ge optimized at the DFT level is a = 5.725 Å. The two Ge(001) surfaces of each slab are passivated with hydrogen atoms to saturate the Ge dangling bonds and remove any localized surface band emerging in the band gap of Ge. A vacuum layer with a thickness of 16 Å is added to separate the neighboring slab images. The Brillouin zone (BZ) has been sampled using an $8 \times 8 \times 1$ Γ -centered k-point grid [58]. The energy gap of the slab has been obtained by calculating the local density of states (LDOS) at the innermost position of the slab, using a $24 \times 24 \times 1$ k-point grid for the 2D BZ, and by taking the energy difference between the highest energy occupied state and the lowest energy unoccupied state in the calculated LDOS. We have adopted the SG15-High combination of norm-conserving pseudopotential and LCAO basis set for germanium. The total energy has been converged to $\sim 10^{-5}$ eV at least. Periodic boundary conditions are imposed in both the in-plane and out-of-plane directions.

For the SGF calculations of the Ge(001) surface, we have attached a semi-infinite bulk region to each of the Ge(001)slabs discussed in the previous paragraph, after removal of the passivating hydrogen atoms on the contacted side. We impose the Dirichlet boundary condition at the boundary located at z = 0 Å between the surface and bulk regions as shown in Fig. 8(a), and the Neumann boundary condition at the boundary located at the distance of 16 Å above the Ge(001) surface in vacuum. All the other computational settings are adopted as for the Ge(001) slab calculations. The LDOS has been calculated at the boundary between the surface region and the bulk electrode, using a 24×24 k-point grid to sample the 2D BZ of the Ge(001) surface, and by taking the energy difference between the highest energy occupied state and the lowest energy unoccupied state in the calculated LDOS.

Note that we have not performed any ion relaxation for neither slab nor SGF model of the Ge(001) surface intentionally, keeping the Ge(001) surface structure the same in both slab and SGF calculations. That allows us to separate the effect of the slab finite size from the effect of the ion relaxation on the band structure of the Ge(001) surface.

b. Results. To compare the electronic structures of the Ge(001) surface calculated with the slab and SGF models, we have done slab (SGF) calculations of the energy gap for the Ge(001) surface as a function of the thickness of the Ge slab (Ge surface region) adopted for modeling of the surface. In



FIG. 6. (a) Local density of states (LDOS) calculated for a slab configuration (blue) or a surface configuration (red) with thickness L = 3a, where the thickness L is given in units of the lattice constant of bulk Ge, a. (b) Same as (a), but for L = 7a. The dashed vertical lines indicate the extremities of the band gap of bulk Ge.

Fig. 6, we show the local density of states (LDOS) calculated at the innermost region of the slab and of surface for two representative thicknesses, L/a = 3 and 7. One can see that the energy gap extracted from the LDOS decreases considerably when increasing the thickness from L/a = 3 to 7, while the energy gap in the innermost region of the slab remains essentially constant and matches the band gap of bulk Ge. In Fig. 7, one can see that the energy gap value for the Ge(001)slab goes slowly to its asymptotic value (which coincides with the band gap of bulk Ge in this case) upon increasing the slab thickness. Contrary to the slab calculations, the energy gap of the Ge(001) surface modeled with the SGF approach is essentially constant across the system, as expected for a surface free from surface states, and does not depend on the value of L. The energy gap of the Ge(001) surface modeled with the SGF approach shows no further dependence on L if the surface region thickness $L/a \ge 4$, whereas there exists a strong thickness dependence of the Ge(001) surface energy gap for the slab model. This suggests that the SGF model of the Ge surface accurately represents the bulk Ge states, eliminating any quantization effects, unlike the Ge slab model where quantization effects are sizable, even for L/a = 7, see Fig. 7.

B. Si film on the Ge(001) surface

In this section, we study a (001)-oriented Si film interfaced with the Ge(001) surface. The main goal of this study is to



FIG. 7. The conduction-band minimum and valence-band maximum calculated for bulk Ge (black dotted lines), the Ge(001) surface (red solid lines, filled circles) modeled with the SGF approach, and the Ge slabs (blue dashed lines, filled diamonds) with different slab thicknesses L given in units of the lattice constant of bulk Ge, a.

show how the band alignment at the Ge(001)|Si interface can be calculated for different doping levels of the Ge substrate, using the SGF approach.

a. Methods. To keep the focus on application of the SGF methodology to the band alignment calculation rather than on understanding of the actual complex structure of the lattice-mismatched Ge(001)|Si interface, we have adopted a simple model to match a Si film on a Ge substrate, where the in-plane lattice parameter of the (minimal) lateral unit cell of the Si film is adjusted to that of the Ge(001) surface. This matching procedure gives rise to the lateral strain of 5.5 % in the Si film. The Si film thickness is chosen to be 2.54 nm. The corresponding Ge(001)|Si heterostructure is illustrated in Fig. 8(a).

We have studied the Ge(001)|Si heterostructure for three different doping levels of the Ge(001) substrate, adopting the atomic compensation charge method for doping the semiconductor structure, see Refs. [37,66] for more details. We have used the SG15-Medium (High) combination of norm-conserving pseudopotential and LCAO basis set for silicon (germanium). All other computational settings are as for the Ge surface calculations in the previous section. In addition, we have done ion relaxation for the top layers of the Ge(001) surface, as well as for the entire Si film in the heterostructure. The forces have been converged to a maximum value of 0.005 eV/Å. The ion relaxation has been allowed in the out-of-plane Z direction only, meaning that the Si film is still strained in the in-plane X and Y directions.

b. Results. The high strain in the supported Si film strongly reduces the energy gap of the corresponding Si slab from 1.28 to 0.46 eV as seen in Fig. 9(b). Interfacing the Si film with the Ge(001) surface gives rise to a charge transfer from the Ge(001) to Si surface. Table I shows the charge (Q_{Si}) induced in the Si film upon formation of the Ge(001)|Si heterostructure for three doping levels of the Ge(001) substrate. Note that Q_{Si} has been scaled with respect to its value at $p = 10^{20} \text{ cm}^{-3}$, Q_{p-Si} . The charge transfer results in electron accumulation on the Si film. The



FIG. 8. (a) The Ge(001)|Si interface structure and the macroscopic in-plane averaged Hartree difference potential, $\langle \delta V_H \rangle$, calculated for the Si film on the *p*-doped Ge substrate ($p = 10^{20} \text{ cm}^{-3}$). The green solid, black dotted and dark magenta solid lines indicate the regions of $\langle \delta V_H \rangle$ corresponding to the Ge(001) surface, the Ge(001)|Si interface, and the Si film, respectively. Note that for visualization purposes the structure has been repeated in the *XY* plane, while only a single lateral unit cell is used for the actual calculations. (b) The macroscopic in-plane averaged Hartree difference potential, $\langle \delta V_H \rangle$, along the *Z* direction for the *p*-doped (blue), and *n*-doped (red, green) Ge substrate.

electron accumulation further increases for increasingly larger n-doping of the Ge(001) substrate.

Figure 8(b) shows the macroscopic in-plane averaged Hartree difference potential $\langle \delta V_{\rm H} \rangle$ [59,64], for the different doping levels. One can see that $\langle \delta V_{\rm H} \rangle$ in the Si film goes upwards with respect to the potential in the bulk Ge region upon increasing the *n*-doping level in the Ge(001) substrate. This behavior of the electrostatic potential is due to the electron transfer from the Ge(001) to Si surface. The corresponding electric field that arises from the negative charge in the Si film penetrates into the Ge(001) substrate. To quantify the band alignment at the interface between the Si film and the Ge(001) substrate, we have calculated the DOS across the heterostructure for each doping level, see Fig. 9. In the case of p-doping, the Ge(001) surface is in the hole accumulation regime, and the charge transfer from the Ge(001) to Si surface results in a short screening length for $\langle \delta V_{\rm H} \rangle$, see Fig. 9(a). When the Ge(001) substrate is *n*-doped, the Ge(001) surface is in the electron depletion regime that gives rise to a longer screening length [see Fig. 9(c)] compared to the case of p doping for comparable magnitudes of the doping level. For a higher *n*-doping level, the screening length gets significantly reduced as seen in Fig. 9(d), meaning that the space charge is confined in the Ge(001) near-surface region. Table I suggests that the charge redistribution across the entire heterostructure gives rise to a doping-dependent potential barrier Φ_b at the Ge(001)|Si interface.

Figure 9 also shows a plot of $\langle \delta V_{\rm H} \rangle$ (overlaid on the DOS) that defines the actual edges of the bands as demonstrated in Ref. [37]. On the Ge side, this is achieved by shifting $\langle \delta V_{\rm H} \rangle$ of an energy equal to the difference between the Fermi energy and the conduction band minimum (CBM) or the valence band maximum (VBM) in bulk Ge. In the silicon thin film, $\langle \delta V_{\rm H} \rangle$ is shifted by an energy equal to the difference between the Fermi energy and the CBM or the VBM in the corresponding silicon slab. Figure 9 allows us to extract the band alignment parameters such as the interface potential, Φ_h , which is given by the distance between the Ge CBM at the interface and in the bulk Ge region. The Φ_b potential acts as a barrier for the electron injection from the Ge(001) substrate into the Si film. Another band alignment parameter of relevance is the conduction (valence) band offset ΔE_c (ΔE_v) that we define as the distance between the conduction band minimum (valence-band maximum) in the bulk Ge region and the surface region of the Si film. A positive sign of the conduction band offset ($\Delta E_c > 0$) indicates that there exists a potential barrier for the electrons propagating from the bulk Ge region to the Si film. The band alignment parameters extracted from the data shown in Fig. 9 are listed in Table I. For each of the three heterostructures with different doping levels, ΔE_v and ΔE_c have the same sign, meaning that the band alignment is of type II with staggered Ge and Si gaps. However, if the conduction and valence band offsets are defined right at the Ge(001)|Si interface, the *p*-doped heterostructure has a type III broken gap. We notice that applying Anderson's electron affinity rule would result in a qualitatively different band diagram for the Ge(001)|Si heterostructure compared to that obtained from the present first-principles study. That suggests that using this empirical rule might not reliably predict the band alignment in complex heterostructures where microscopic details of the interfaces between dissimilar semiconducting materials matter.

Figure 9 also suggests that there exist Ge(001) states that penetrate into the Si film, and this state penetration is related to one of the mechanisms responsible for the electron donation to the Si film, in agreement with earlier predictions for semiconductor heterojunctions [67]. This is particularly evident for the highly *n*-doped Ge(001) substrate as shown in Fig. 9(d), where we see that the $\langle \delta V_{\rm H} \rangle$ potential in the Si film virtually follows the DOS penetration profile related to the conduction band states of the near Ge(001) surface region.

Using Fig. 9, one can conclude that some Si states also penetrate into the Ge(001) substrate. In particular, for the highly *n*-doped Ge(001) substrate this gives rise to a nonmonotonic behavior of the $\langle \delta V_{\rm H} \rangle$ potential, which has a minimum near the interface. Note that there exist no midgap energy levels at the surface of the Si film as seen in Fig. 9, confirming that hydrogen passivation of the Si(001) surface has efficiently removed all the surface point defects related to the Si dangling bonds. Similarly, we do not find any localized interface states at the Ge(001)|Si interface. All the states at the interface arise from penetration of either Ge substrate or Si film states across the interface.

In conclusion, the present study demonstrated that the SGF approach provides an insightful, accurate, computationally



FIG. 9. Band diagrams showing the density of states (DOS) across the Ge(001)|Si heterostructure for (a) *p*-doped Ge(001) substrate with $p = 10^{20}$ cm⁻³, and *n*-doped Ge(001) substrate with (c) $n = 10^{20}$ cm⁻³ and (d) 10^{21} cm⁻³. Notice that $\langle \delta V_H \rangle$ in the Si region has been shifted to match the energy gap of the Si film. (b) Band diagram of the unsupported Si film (Si slab) showing the $\langle \delta V_H \rangle$ superimposed on the DOS.

efficient way for calculation and analysis of complex semiconductor heterostructures at the microscopic level within the framework of DFT. We have shown that the SGF approach is superior compared to the commonly used slab approach as it accounts for bulk states of semiconductor substrates in an exact manner, unlike the slab approach that suffers from finite-size effects.

V. SURFACE STATES

Electronic surface states are notoriously difficult to describe using the slab method, as equivalent states localized on both

TABLE I. Band alignment parameters of the Ge(001)|Si interface for the three different doping levels of the Ge(001) substrate. Q_{Si} (Q_{p-Si}) is the induced charge in the (*p*-doped) Si film, where $Q_{p-Si} =$ $0.49 \times 10^{12} e/cm^2$. Φ_b is the Schottky barrier at the Ge(001)|Si interface. ΔE_v (ΔE_c) is the offset between the Si film and bulk Ge valence (conduction) band maxima (minima). Note that the Q_{Si} charge is calculated by integrating the electron difference density over the Si film [59].

Doping level cm ⁻³	$Q_{\mathrm{Si}} - Q_{p-\mathrm{Si}} = e/\mathrm{cm}^2$	Φ_b eV	ΔE_v eV	ΔE_c eV
$p = 10^{20}$	0	0.08	-0.41	-0.63
$n = 10^{20}$	-2.44×10^{12}	0.39	0.28	0.08
$n = 10^{21}$	-6.70×10^{12}	0.15	0.50	0.27

surfaces of the slab will interact strongly if the slab is not thick enough. *A posteriori* corrections [68] are then needed to decouple the surface states and to correctly model the field dependence of the surface state properties. In this section, we show that the surface states are naturally taken into account within the framework of the SGF method, which deals with a single surface only, and external fields can be applied by shifting the potential near the surface in the vacuum in a simple manner.

We focus here on studying the Shockley surface state that is present at the center of the Brillouin zone on the (111) surfaces of noble metals [69], and the topologically protected surface states that are present at the surface of topological insulators (TIs) [70,71]. As an example of a Shockley-type surface state, we consider the Ag(111) surface, for which accurate experimental data from photoemission spectroscopy (PES) and scanning tunneling spectroscopy (STS) are available. It has also been demonstrated that external electric fields can alter the surface state and change its overall properties [72,73]. As a prototypical TI surface, we consider a Se-terminated Bi₂Se₃(111) surface. A previous work has also adopted a SGF-type approach to describe the formation of surface states on the $Bi_2Se_3(111)$ surface, but that approach was based on a parametrized effective Hamiltonian [70,74]. In the following, we give a first principles, atomistic description of the surface states that is not based on any adjustable parameters. For both the Ag(111) and $Bi_2Se_3(111)$ surfaces, the surface states are identified in the surface band structure, which has been described with the density of states (DOS) calculated along the $M \to \Gamma \to K$ k-point path in the 2D Brillouin zone (BZ) of the surface.

A. Ag(111) Shockley surface state

a. Methods. We have done the ATK-SGF calculations using a surface region comprised of 27 atomic monolayers and a vacuum layer with a thickness of 20 Å. This large number of Ag(111) monolayers is used to increase the contribution of the bulk Ag states to the electronic structure of the surface region projected onto the 2D BZ of the (111) surface. We notice that the Shockley surface state is highly localized at the surface, and it can be accurately described by using just seven atomic monolayers in the surface region. It means that we could adopt, in principle, a smaller surface region, adding the DOS of bulk Ag to the SGF-calculated DOS of the surface region.

The surface 2D BZ has been sampled using a 21×21 kpoint grid. The corresponding 3D BZ in the bulk electrode has been sampled using $21 \times 21 \times 201$ k points. Following the procedure described in Ref. [65], we have included a layer of ghost atoms above the top monolayer of the surface to accurately describe the decay of the surface electron density into the vacuum. The SGF surface calculations have been done for different external electric fields applied perpendicularly to the surface plane, ranging from $E_z = -0.27$ V/Å to +0.27 V/Å, at a regular step of $\Delta E_z = 0.054$ V/Å. In the SGF method, an electric field is imposed with the Dirichlet boundary condition by shifting the electrostatic potential value in the vacuum, while keeping the chemical potential of the semi-infinite bulk region unchanged. Note that this procedure resembles an experimental measurement in which the surface is exposed to an external field generated by a scanning tunneling microscopy tip [72]. The Ag(111) surface structure shown in Fig. 10 has been built using the DFT-PBE calculated lattice constant of bulk Ag ($a_{Ag} = 4.086$ Å). Subsequently, we have done ion relaxation for the top surface layers. More information on the computational details of the ATK-SGF calculations can be found in Sec. II E.

b. Results. Figure 10 shows the difference $(\Delta V_{\rm H})$ in the Hartree potential induced by the external electric field,

$$\Delta V_{\rm H} = V_{\rm H} - V_{\rm H}^0,\tag{11}$$

where $V_{\rm H}^0$ is the Hartree potential at zero field. Figure 10 suggests that applying the external field induces a perturbation of the Hartree potential that is far beyond the Ag(111) topmost monolayer, located at $Z_{\rm surf} = 68.64$ Å. For the electric field magnitude of $E_z = \pm 0.27$ V/Å, the oscillations of the $\Delta V_{\rm H}$ potential at $Z < Z_{\rm surf}$, which are clearly seen in Fig. 10(b), indicate that the field-induced perturbation of the surface electronic structure is completely screened after the seventh innermost Ag(111) monolayer only, with the screening being somewhat more efficient for positive than for negative biases.

The 2D surface electronic band structure of the Ag(111) surface is shown in Fig. 11(a) for $E_z = 0$ (green) and ± 0.27 V/Å (violet and red). The bottom of the surface state band (indicated by solid lines) is located at the Γ point, above the highest occupied bulk state (indicated by dashed lines). At zero field, the energy at the bottom of the surface state band is $\varepsilon - \varepsilon_F = -147$ meV, in good agreement with the value of -120 ± 1 meV obtained from PES measurements [75]. A fit



FIG. 10. (a) Field-induced difference in the macroscopic in-plane averaged Hartree potential in the Ag(111) surface region, $\Delta V_{\rm H}$, plotted in the out-of-surface-plane (Z) direction, for different external electric fields, E_z . The interface between the semi-infinite bulk Ag region and surface region is at Z = 0 Å. The black dashed vertical line at $Z_{\rm surf} = 64.68$ Å indicates the position of the Ag(111) top monolayer. Ag atoms are shown as gray spheres. (b) 100 × zoom of the inset (a) for $E_z = \pm 0.27$ V/Å.

of the parabolic dispersion of the surface state band using the free-electron gas model, $\varepsilon = (\hbar k)^2 / 2m^*$, results in a value for the electron effective mass of $m^* = 0.306 m_e$, in close agreement with the value $m^* = 0.31 \pm 0.01 m_e$ measured with STS [76].

Applying an electric field gives rise to a linear Stark shift $(\Delta \varepsilon)$ of the surface state energy, which follows the sign of the applied field. This behavior is clearly seen in Fig. 11(a), and is consistent with several experimental [72,73] and theoretical reports [68]. Figure 11(b) shows how the Stark shift computed for the Ag(111) surface states depends on the external electric field. A linear fit to the $\Delta \varepsilon$ versus E_z data in Fig. 11(b) yields a slope of $8 \times 10^{-3} e \cdot \text{Å}$. It is evident that the field alters the dispersion of the surface state bands, resulting in a variation of the electron effective mass from $m^* = 0.301 m_e$ to $0.314 m_e$ corresponding to $E_z = -0.27$ and +0.27 V/Å, respectively. This is in agreement with the results previously reported for the Cu(111) surface state [68].

Strikingly, the variation of the Stark shift, $\Delta \varepsilon$, is linear with respect to the E_z even in the limit of a vanishing field, when the shift calculated with the slab model would exhibit an avoided crossing behavior as a result of the interaction between the surface states at the two surfaces of the slab [68]. Figure 11(b) also shows that the position of the lower band edge (LBE) of the bulk bands remains fixed in the SGF-calculated band structure of the Ag(111) surface, while E_z changes. We notice that this physically correct behavior is not captured by the slab model, as the Ag states of the thin slab structure are not pinned to the



FIG. 11. (a) Band structure of the Ag(111) surface along the M \rightarrow $\Gamma \rightarrow K k$ -point path of the 2D Brillouin zone in the vicinity of the Fermi energy and close to the Γ point, for applied fields of -0.27 V/Å (left inset), 0.0 V/Å (center inset) and +0.27 V/Å (right inset). The horizontal solid (dashed) lines indicate the lowest energy of the surface states (highest occupied bulk states) at the Γ point. (b) Stark energy shift ($\Delta \varepsilon$) of the Ag(111) surface state against the E_z -electric field applied. Filled circles and black solid line correspond to the surface state (SS) and a linear fit of the calculated data, respectively. The filled squares correspond to lower band edge (LBE).

true bulk Ag states, and therefore the corresponding bulk-like bands of the slab can be shifted by the applied electric field.

B. Bi₂Se₃(111) topologically protected surface state

a. Methods. To study the topologically protected surface states on the $Bi_2Se_3(111)$ surface, we have constructed the $Bi_2Se_3(111)$ surface structure, using a fully relaxed bulk Bi_2Se_3 unit cell, where the forces and stress were converged to 0.05 eV/Å and 1 GPa, respectively. The surface region comprises 37 atomic monolayers, corresponding to 7.6 quintuple layers (QLs). The principal layer of the bulk region consists of 3 QLs. For the sake of simplicity, we have not done ion relaxation of the surface. A noncollinear spin formalism including spin-orbit



FIG. 12. (a) Band structure of the Se-terminated Bi₂Se₃(111) surface along the $M \rightarrow \Gamma \rightarrow K$ *k*-point path in the 2D Brillouin zone in the vicinity of the Fermi energy, close to the Γ point. (b) Density of states at the Γ point. The red horizontal lines in both insets indicate the position of the Fermi energy.

coupling has been employed in all the ATK-SGF calculations of the topologically protected surface states [41]. The 2D (3D) BZ of the surface (bulk) region has been sampled with a 9×9 ($9 \times 9 \times 201$) *k*-point grid, and the broadening of the Fermi–Dirac distribution for calculating the electron occupation has been set to a rather small value of ~0.004 eV.

b. Results. In Fig. 12(a), one can see the electronic band structure of the $Bi_2Se_3(111)$ surface, calculated with the ATK-SGF method. This figure suggests that there exist two topologically protected surface states inside the electronic gap of bulk Bi_2Se_3 , as the two surface states cross at the Fermi energy (the Dirac point), around which the dispersion is essentially linear. This is in agreement with previous work where either the slab [41] or SGF approach [70,74] have been adopted to study the topologically protected surface states.

By examining the ATK-SGF calculated surface DOS at the Γ point [see Fig. 12(b)], we find that the electronic energy gap of the bulk material is of 250 meV, and the conduction band minimum is at 170 meV above the Dirac point associated with the surface states, in good agreement with the values reported in earlier angle-resolved PES measurements on Bi₂Se₃ single crystals [71]. Importantly, a single narrow peak is present at the Fermi energy, which is related to the spin-degenerate state arising from the intersection between the two spin-locked surface states. The peak has a Lorentzian shape with a width given by the actual numerical value of the infinitesimal, δ , used for computing the Green's function of the surface region in Eq. (5). This degeneracy between the two surface bands arises naturally within the framework of the SGF formalism, whereas in finite-size slab models of the topological insulator surfaces, the interaction between evanescent states localized at the two surfaces of the slab leads to an unphysical energy gap opening that is inversely proportional to the slab thickness [77]. This allows us to conclude that the SGF method provides an accurate description of the topologically protected surface states, compared to the slab method.

VI. SURFACE CHEMISTRY IN EXTERNAL ELECTROSTATIC FIELDS

The properties of adsorbed species at electrochemical metal-solution interfaces depend on the applied electrode potential and hence the electric field. Several theoretical works have considered the response of chemisorption binding energies and vibrational frequencies to a potential bias, using either slab calculations or metallic clusters [78,79]. In particular, Bonnet and co-workers have recently used the slab model in combination with the effective screening medium [80] method to investigate the vibrational response of carbon monoxide on a platinum electrode from first principles [81]. Other works have focused on the very high fields needed to rip atoms out of the surface during field emission processes [82,83].

A slab is a confined system in the out-of-plane direction, so any charging of adsorbates on the slab surface must be counterbalanced by an opposite charge in the slab, altering the electron chemical potential of the finite-size slab system. We notice that no change of the chemical potential would take place in a truly semi-infinite surface system. In the SGF approach, the chemical potential of the surface with adsorbates is fixed by an infinite reservoir of electrons (bulk region) coupled to the surface region. The electrons are allowed to be transferred between the surface and bulk regions in a fully self-consistent manner. The adsorbates may therefore be charged with the charges that originate from the bulk region without altering the chemical potential of the surface system, unlike the slab system.

We here consider atomic iodine adsorbed on the Pt(111) surface [84,85], which is a system of relevance for dyesensitized solar cells [86]. We show that the SGF method is a natural choice for studying the chemical properties of adsorbates on the surface in an external electrostatic field, as it allows charging of the iodine atom from the electron reservoir, instead of the limited electron supply in a slab system.

a. Methods. We have constructed a 2×2 Pt(111) surface, using a crystal structure of bulk Pt with the DFT-optimized lattice parameter, $a_{Pt} = 3.956$ Å. We have adopted three atomic (111) monolayers for the principal layer of the bulk region, and nine atomic (111) monolayers for the central region. The vacuum thickness has been set to 20 Å. The Neumann boundary condition has been imposed in the vacuum region. The 2D Brillouin zone has been sampled using a 6×6 *k*-point grid. The top six monolayers of the Pt(111) surface have been relaxed within the framework of the DFT approach, see Sec. II E for more computational details. For surface calculations with adsorbates, atomic iodine has been placed in the fcc hollow site, and the iodine atom and top six monolayers of the Pt(111) surface have been relaxed, converging forces to 0.05 eV/Å.

To calculate the equilibrium separation distance between the iodine atoms in a single I₂ molecule, we have adopted a large unit cell with the sufficiently thick vacuum padding around the molecule to avoid iteration between the repeating images. Γ -only *k* points sampling and 4 meV broadening of the Fermi-Dirac distribution are used for this calculation, yielding an I–I equilibrium bond length of 2.73 Å.

The potential-energy profile for the interaction of a single iodine atom with the Pt(111) surface has been calculated by

displacing the I atom away from its equilibrium position on the surface along the surface normal in steps of $\delta d = 0.25$ Å. For a given Pt(111)–I separation distance *d*, the energy of the system has been calculated by using the grand canonical potential, as defined for an open system coupled to an electron reservoir,

$$\Omega[\rho] = E[\rho] - e\,\delta n_{\text{bulk}}\,\mu_{\text{bulk}},\tag{12}$$

where *E* is the total energy of the surface region, ρ is the electronic density in the surface region, and δn_{bulk} is the number of electrons exchanged with the electron reservoir with chemical potential μ_{bulk} . The adsorption energy $\Delta \Omega$ is then evaluated as

$$\Delta\Omega(d) = \Omega_{\text{I/Pt}(111)}(d) - \Omega_{\text{Pt}(111)} - \frac{1}{2}E_{\text{I}_2} - \Delta_{\text{CP}}(d), \quad (13)$$

where $\Omega_{I/Pt(111)}$ and $\Omega_{Pt(111)}$ are the grand canonical potentials of the Pt(111) system with and without adsorbate, respectively. E_{I_2} is the total energy of a I₂ molecule, which is equivalent to Ω_{I_2} since an isolated molecule cannot exchange particles with a reservoir. The counterpoise (CP) correction $\Delta_{CP}(d)$ is similar in spirit to the standard Boys-Bernardi CP correction to account for the basis set superposition error [88],

$$\Delta_{\rm CP}(d) = (\Omega_{\rm I^*/Pt(111)}(d) - \Omega_{\rm Pt(111)}) + (E_{\rm I/Pt(111)^*}(d) - E_{\rm I}) + \frac{1}{2} (E_{\rm I_2}^* - E_{\rm I_2}),$$
(14)

where $E_{I_2}^*$ is the total energy of a fictitious I_2 molecule in which one of the two iodine atoms is assumed to be a ghost atom, $\Omega_{I^*/Pt(111)}$ is the grand canonical potential of the I/Pt(111) surface in which the iodine atom is treated as a ghost atom, and $E_{I/Pt(111)^*}$ is the total energy of the corresponding I/Pt(111) slab in which the platinum atoms are treated as ghost atoms.

Field-dependent potential profiles have been obtained by imposing the Dirichlet boundary condition with different electrostatic potential values in the vacuum region, corresponding to external electric fields in the range from -6 to +6 V/Å. We notice that, due to the use of a LCAO basis set, the present approach is not suitable for describing field-emission processes, in which electrons are moved from the surface to the vacuum, due to an applied electrical field [89].

b. Results. Figure 13 shows the potential profile calculated using the DFT-SGF method for an iodine atom interacting with the Pt(111) surface. The equilibrium adsorption energy obtained from this profile agrees well with the measured and plane-wave DFT-calculated energies [87]. The Pt(111)–I separation distance, d, is defined with respect to the top monolayer of the Pt(111) surface, and the equilibrium separation distance d_0 , obtained with the DFT-SGF method is of 2.28 Å.

Figure 14(a) suggests that applying an external electrostatic field has a significant impact on the SGF-calculated potential profile for fields in the range from -6 to +6 V/Å. The potential-energy profile in the vacuum region is pushed down for increasingly large negative fields, lowering the energy barrier for desorption, whereas positive fields have the opposite effect. This behavior is opposite compared to that found for field-induced desorption of Na [78] and Al [83] adatoms on Al(111), and can be ascribed to the propensity of the iodine adatom to form a stable anion, rather than a stable cation, due to its halogenic character. The equilibrium I–Pt(111) separation distance d_0 is also affected by the electrostatic field change,



FIG. 13. Zero-field potential-energy curve (solid lines, filled circles) for atomic iodine adsorbed on the Pt(111) surface, calculated using the SGF method. Cubic interpolation of the PEC near the minimum yields an adsorption energy of -1.45 eV at an equilibrium I–Pt(111) separation distance (d_0) of 2.28 Å. The measured and plane-wave DFT-calculated adsorption energies (dotted lines) are taken from Ref. [87].

as illustrated in Fig. 14(b). In this figure, the Mulliken charge on the iodine atom is shown as function of the applied field



FIG. 14. (a) Finite-field potential-energy curves for atomic iodine adsorbed on the Pt(111) surface, for electrostatic fields ranging from +6 V/Å (red) over zero field (green) to -6 V/Å (black). For all the curves, zero energy is chosen at the equilibrium position of the adsorbate. (b) Equilibrium I–Pt(111) separation distance d_0 (solid line) and Mulliken population (dashed line) on the iodine atom as a function of the applied field. Positive Mulliken population means accumulation of electrons on the iodine atom.

strength. As the field turns more negative, electron charge accumulates on the iodine atom, and the I–Pt(111) separation distance increases due to the larger anionic character of the adatom. As the I–Pt(111) distance is increased, the Mulliken population on the iodine atom remains essentially constant for negative applied fields, whereas it becomes increasingly more negative for positive values of the applied field, reaching values of -0.37 ($E_z = +3$ V) and -0.85 ($E_z = +6$ V) at d = 5 Å. In conclusion, we notice that this charge accumulation on the iodine atom does not require a corresponding charge of opposite sign in the near-surface region, as it is taken from the semi-infinite bulk region instead. That shows a crucial difference between the traditional slab and Green's-function approaches for surface chemistry calculations.

VII. CONCLUSIONS

In this work, we presented the state-of-the-art implementation of the Green's function formalism [16-22] for accurate first-principles simulations of surfaces within the framework of density functional theory. Unlike the slab model that is traditionally used in computational surface science, the Green's-function approach allowed us to model the surface as a truly semi-infinite system by coupling a surface region to an electron reservoir. We were able to do first-principles calculations of surface systems that are free from the drawbacks present in the slab calculations, which are affected by finite-size effects. Furthermore, the computational cost of the Green's-function based surface calculations was shown to have a linear scaling with respect to the length of the surface region. For large systems, it provides a better alternative to the slab calculations that have a cubic scaling with respect to the slab thickness.

Using the Green's-function approach was shown to improve the accuracy of both quantitative and qualitative description of surface properties that are notoriously difficult to address using the slab approach, including metal work functions, surface states of metals and topological insulators, and energy gaps of semiconductor surfaces. We demonstrated the actual advantages of using Green's functions for several advanced physics and chemistry studies of surfaces. The adopted approach allowed us to accurately calculate the work functions of several transition metal surfaces. We found that the first-principles Green's-function approach combined with the analysis of physical properties based on the projected density of states and Hartree difference potential makes possible to quantitatively determine the band diagram across semiconductor heterostructures such as an ultrathin Si film on a doped Ge substrate in atomistic simulations. We found that it is crucial to adopt the surface Green's-function method for correct description of topologically protected states in the Bi₂Se₃ topological insulator, as well as the effect of an external electric field on the surface state of the Ag(111)surface. The charge transfer effects for metal surfaces with adsorbates such as iodine atoms on the Pt(111) surface, turned out to be naturally captured within the framework of the Green's-function formalism that allows describing the surface structures coupled to an electron reservoir.

In conclusion, the present results suggested that the Green's-function approach to surface calculations is a superior

tool compared to more traditional approaches to surface modeling. Given the demonstrated advantages of this approach, in this work, we showed how one may increase the accuracy of DFT-based surface calculations, and how the applicability of first-principle, atomistic modeling can be extended towards challenging problems in surface science.

APPENDIX: PSEUDOPOTENTIALS AND BASIS SETS

The accuracy of the DFT calculations based on the pseudopotential LCAO approach depends on the choice of pseudopotentials and basis sets. We employ norm-conserving pseudopotentials in the Kleinman–Bylander form [90]. The basis functions are atom-centered orbitals constructed by solving the Schrödinger equation for a single atom in a confinement potential [39,91].

We have implemented in the ATK high-accuracy pseudopotentials and localized basis sets for all elements up to Z = 83(Bi), excluding lanthanides. We have used the SG15 suite of optimized norm-conserving Vanderbilt pseudpotentials from Ref. [40]. For a number of chemical elements, we have improved the pseudopotential quality by adding a nonlinear core correction [92]. Both scalar-relativistic and fully relativistic versions of all the pseudopotentials are available in the ATK software package. The fully relativistic pseudopotentials allow for DFT calculations with spin-orbit coupling included [93].

To construct high-accuracy LCAO basis sets, we have first taken a large set of pseudo-atomic orbitals similar to the "tight tier 2" basis sets used in the FHI-aims package [91]. These basis sets typically have five orbitals per pseudopotential valence electron, a range of 5 Å for all orbitals, and include angular momentum channels up to l = 5. We find that such a large LCAO basis set gives essentially the same computational accuracy as fully converged plane-wave calculations. We have then reduced the range of the orbitals by requiring that the overlap of the contracted wave function must change less than 0.1% with the original wave function. Such a reduction of the orbital range decreases the number of matrix elements that needs to be evaluated, and at the same time this does not alter the accuracy of LCAO calculations. In the following, this LCAO basis set will be called *Ultra*.

From the Ultra basis set we generate two reduced basis sets, *High* and *Medium*. The High basis set is generated by reducing the number of basis set orbitals such that the DFT total energy of suitably chosen test systems does not change by more than 1 meV (per atom). For each element, the test set consists of the element in its experimental (300 K) bulk structure at different lattice constants (that allows for computing the Δ value [94,95]), and dimers and octamers of the element at different inter-atomic distances. We see from Table II that the

TABLE II. Δ values calculated with different basis sets. The plane-wave (PW) value is obtained using QUANTUM ESPRESSO with the same SG15 pseudopotentials [95].

TABLE III. Summary of ATK-LCAO calculations for rock salts and perovskites test sets. The shown RMS errors are calculated relative to all-electron calculations. The test sets and the VASP results are taken from Ref. [96].

	Medium	High	Ultra	VASP
Rock salt latt. const. (%)	0.40	0.24	0.23	0.15
Perovskite latt. const. (%)	0.36	0.24	0.18	0.13

Ultra and High basis sets have essentially the same Δ value, indicating that they are equally accurate. The Medium basis set is constructed by further reduction of the High basis set, while keeping the Δ value below 4 meV.

In order to further validate the constructed SG15 pseudopotentials and basis sets, we have performed benchmark calculations for rock salt and perovskite crystals, as described in Ref. [96]. For each bulk structure, the equation of state is calculated at fixed internal coordinates, and the equilibrium lattice constant and bulk modulus are then computed. Results are benchmarked against the scalar-relativistic all-electron calculations [96]. Table III shows the root-mean-square (RMS) deviations from the all-electron reference for the calculated lattice constants and bulk moduli. For the sake of comparison, statistics for plane-wave VASP [31] calculations is also included in this table. We see that the accuracy of the DFT calculations done with the SG15 pseudopotentials and High (or Ultra) LCAO basis sets is comparable to that of plane-wave calculations, while the use of the Medium basis sets gives a slightly larger deviation from all-electron results.

We find that for typical atomistic simulations with less than 500 atoms, the LCAO calculations done with the Medium basis set is twice as fast as that done with the High basis set, which allows for 10 times faster LCAO calculations compared to the ones done the Ultra basis set. Furthermore, using the Medium basis set typically permits one to do LCAO calculations an order of magnitude faster than plane-wave calculations. In summary, the Ultra basis set enables essentially the same accuracy of LCAO-based DFT calculations as plane-wave calculations, at similar cost for typical 200-atom systems. Using the Medium and High basis sets gives somewhat less accurate results of LCAO-based DFT calculations, allowing for an order of magnitude speedup.

1. Accurate semiconductor band gaps

It is known that density functionals based on local density approximation (LDA) and generalized gradient approximation (GGA) do not allow for an accurate calculation of energy band

TABLE IV. Empirical pseudopotential projector shifts employed in the PPS-PBE method with the SG15-Medium combination of pseudopotential and basis set. The shifts α_s , α_p , and α_d are applied to *s*, *p*, and *d* orbitals, respectively.

						0
	Medium	Hıgh	Ultra	PW	Si	+21.
Δ (meV)	3.45	1.88	2.03	1.3	Ge	+13.

	$lpha_s$	α_p	$lpha_d$
Si	+21.33 eV	-1.43 eV	
Ge	+13.79 eV	+0.22 eV	-2.03 eV



FIG. 15. Band structures of bulk Si and Ge, calculated with the PPS-PBE method. Note that the fundamental Γ -*L* band gap of bulk Ge calculated using the PBE functional is zero, in contradiction with experiment, whereas the PPS-PBE calculated band gap and other band energies agree with the experimental data in a semi-quantitative manner. The fundamental gaps for bulk Si and Ge are $E_g^{Si} = 1.14 \text{ eV}$ and $E_g^{Ge} = 0.64 \text{ eV}$, respectively.

TABLE V. Material parameters computed for Si, Ge, and SiGe calculated using ATK and the PBE (first column), and PPS-PBE (third column) approaches. The second column shows the corresponding HSE values calculated using VASP and taken from Ref. [97]. The experimental reference values, taken from Refs. [97–99], are shown in the fourth column.

	PBE	HSE ^a	PPS-PBE	Exp. ^b
Si				
a (Å)	5.468	5.435	5.443	5.430
B (GPa)	90.4	97.7	102.5	100.8
$E_{\rm g}$ (eV)	0.58	1.14	1.10	1.17
$m^*_{\Delta}(m_{\rm e})$	0.19		0.22	0.19
Ge				
a (Å)	5.815	5.682	5.735	5.658
B (GPa)	60.9	71.3	67.4	77.3
E_{g} (eV)	0.00	0.72	0.69	0.74
m_L^* (m_e)	0.09		0.14	0.08
SiGe				
a (Å)	5.62		5.56	5.54
B (GPa)	72.0		95.3	86.5
$E_{\rm g}$ (eV)	0.69		0.89	0.97
$m^*(m_e)$	0.20		0.22	0.19

^aVASP, Ref. [97].

^bFrom Refs. [97–99].

the combination of SG15-pseudopotentials and Medium basis sets described in the previous section. The corresponding PPS parameters are listed in Table IV. Figure 15 shows the PPS-PBE calculated band structures of bulk Si and Ge. We find that shifting the pseudopotential projectors allows us to significantly improve not only the band gap values (compared to experiment), but also other band energies corresponding to higher conduction band valleys in Si, Ge, and related alloys.

Furthermore, Table V compares the material parameters computed with the PPS-PBE and standard PBE approaches for Si, Ge, and a SiGe alloy to the material parameters obtained with the computationally more expensive HSE hybrid functional [100]. The PPS-PBE calculated lattice constants, bulk moduli, and fundamental band gaps are in significantly better agreement with experimentally measured material parameters than the parameters calculated with the PBE approach, and are on par with HSE predictions, in general.

gaps of semiconductors [47]. To overcome this issue we have introduced empirical shifts of the nonlocal projectors in the SG15 pseudopotentials, in spirit of empirical pseudopotentials proposed by Zunger and co-workers [48]. The pseudopotential projector shifts (PPS) have been adjusted to reproduce technologically important properties of semiconductors such as the fundamental band gap and lattice constant. In the PPS method, the nonlocal part of the pseudopotential, \hat{V}_{nl} , is modified in the following way:

$$\hat{V}_{\rm nl} \to \hat{V}_{\rm nl} + \sum_{l} |p_l\rangle \alpha_l \langle p_l|,$$
 (A1)

where the sum is over all projectors p_l , and α_l is an empirical parameter that depends on orbital angular momentum quantum number, l. We note that this approach does not increase the computational cost of DFT calculations.

We have applied the PPS method in calculations for Si, Ge, and SiGe alloys, using the GGA-PBE functional [57] and

- P. Fulde, *Electron Correlations in Molecules and Solids* (Springer, Berlin, 1995).
- [2] G. Ceder, Y.-M. Chiang, D. R. Sadoway, M. K. Aydinol, Y.-I. Jang, and B. Huang, Nature (London) **392**, 694 (1998).
- [3] P. Zhang, V. H. Crespi, E. Chang, S. G. Louie, and M. L. Cohen, Nature (London) 409, 69 (2001).
- [4] J. K. Nørskov, M. Scheffler, and H. Toulhoat, MRS Bull. 31, 669 (2006).
- [5] C. Wood and D. Jena (eds.), Polarization Effects in Semiconductors: From Ab Initio Theory to Device Applications (Springer, New York, 2008).
- [6] A. Jain, S. P. Ong, G. Hautier, W. Chen, W. D. Richards, S. Dacek, S. Cholia, D. Gunter, D. Skinner, G. Ceder, and K. A. Persson, APL Mater. 1, 011002 (2013).
- [7] C. E. Lekka, M. J. Mehl, N. Bernstein, and D. A. Papaconstantopoulos, Phys. Rev. B 68, 035422 (2003).
- [8] N. Martsinovich, D. R. Jones, and A. Troisi, J. Phys. Chem. C 114, 22659 (2010).
- [9] N. E. Singh-Miller and N. Marzari, Phys. Rev. B 80, 235407 (2009).
- [10] C. J. Fall, N. Binggeli, and A. Baldereschi, J. Phys.: Condens. Matter 11, 2689 (1999).

- [11] D. Stradi, S. Barja, C. Díaz, M. Garnica, B. Borca, J. J. Hinarejos, D. Sanchez-Portal, M. Alcamí, A. Arnau, A. L. Vazquez de Parga, R. Miranda, and F. Martín, Phys. Rev. B 88, 245401 (2013).
- [12] G. Ali Shah, M. W. Radny, P. V. Smith, and S. R. Schofield, J. Phys. Chem. C 116, 6615 (2012).
- [13] K. Sagisaka, J. Nara, and D. Bowler, J. Phys.: Condens. Matter 29, 145502 (2017).
- [14] G. te Velde and E. J. Baerends, Chem. Phys. 177, 399 (1993).
- [15] D. F. Tracey, B. Belley, D. R. McKenzie, and O. Warschkow, AIP Adv. 3, 042117 (2013).
- [16] J. E. Inglesfield and G. A. Benesh, Phys. Rev. B 37, 6682 (1988).
- [17] J. M. MacLaren, S. Crampin, D. D. Vvedensky, and J. B. Pendry, Phys. Rev. B 40, 12164 (1989).
- [18] H. L. Skriver and N. M. Rosengaard, Phys. Rev. B 43, 9538 (1991).
- [19] J. Kudrnovský, I. Turek, V. Drchal, P. Weinberger, N. E. Christensen, and S. K. Bose, Phys. Rev. B 46, 4222 (1992).
- [20] L. Szunyogh, B. Újfalussy, P. Weinberger, and J. Kollár, Phys. Rev. B 49, 2721 (1994).
- [21] H. Ishida, Phys. Rev. B 63, 165409 (2001).
- [22] N. Papior, N. Lorente, T. Frederiksen, A. García, and M. Brandbyge, Comput. Phys. Commun. 212, 8 (2017).
- [23] S. N. P. Wissing, C. Eibl, A. Zumbülte, A. B. Schmidt, J. Braun, J. Minár, H. Ebert, and M. Donath, New J. Phys. 15, 105001 (2013).
- [24] M. Brandbyge, J.-L. Mozos, P. Ordejón, J. Taylor, and K. Stokbro, Phys. Rev. B 65, 165401 (2002).
- [25] T. Ozaki, K. Nishio, and H. Kino, Phys. Rev. B 81, 035116 (2010).
- [26] S. Datta, *Electronic Transport in Mesoscopic Systems* (Cambridge University Press, Cambridge, 1997).
- [27] P. Hohenberg and W. Kohn, Phy. Rev. 136, B864 (1964).
- [28] W. Kohn and L. J. Sham, Phys. Rev. 140, A1133 (1965).
- [29] M. C. Payne, M. P. Teter, D. C. Allan, T. A. Arias, and J. D. Joannopoulos, Rev. Mod. Phys. 64, 1045 (1992).
- [30] B. Hammer and J. K. Nørskov, Adv. Catal. 45, 71 (2000).
- [31] G. Kresse and J. Furthmüller, Phys. Rev. B 54, 11169 (1996).
- [32] R. Cuadrado and J. I. Cerda, J. Phys.: Condens. Matter 24, 086005 (2012).
- [33] H. Ishida, Phys. Rev. B 90, 235422 (2014).
- [34] T. Olsen, Phys. Rev. B 94, 235106 (2016).
- [35] J. E. Inglesfield, *The Embedding Method for Electronic Structure* (IOP, Bristol, 2015).
- [36] N. Marzari, Rev. Mod. Phys. 84, 1419 (2012).
- [37] D. Stradi, U. Martinez, A. Blom, M. Brandbyge, and K. Stokbro, Phys. Rev. B 93, 155302 (2016).
- [38] ATOMISTIX TOOLKIT version 2017.2 Synopsys QuantumWise (www.quantumwise.com). QuantumWise A/S is now part of Synopsys, and from the upcoming version ATK will be part of the QuantumATK suite.
- [39] J. M. Soler, E. Artacho, J. D. Gale, A. García, J. Junquera, P. Ordejón, and D. Sánchez-Portal, J. Phys.: Condens. Matter 14, 2745 (2002).
- [40] M. Schlipf and F. Gygi, Comput. Phys. Commun. 196, 36 (2015).
- [41] P.-H. Chang, T. Markussen, S. Smidstrup, K. Stokbro, and B. K. Nikolić, Phys. Rev. B 92, 201406 (2015).

- [42] W. Kohn, A. D. Becke, and R. G. Parr, J. Phys. Chem. 100, 12974 (1996).
- [43] R. G. Parr and W. Yang, *Density-Functional Theory of Atoms and Molecules*, International Series of Monographs on Chemistry (Oxford University Press, Oxford, 1994).
- [44] D. R. Hamann, M. Schlüter, and C. Chiang, Phys. Rev. Lett. 43, 1494 (1979).
- [45] J. Junquera, O. Paz, D. Sánchez-Portal, and E. Artacho, Phys. Rev. B 64, 235111 (2001).
- [46] T. Ozaki, Phys. Rev. B 67, 155108 (2003).
- [47] P. Mori-Sánchez, A. J. Cohen, and W. Yang, Phys. Rev. Lett. 100, 146401 (2008).
- [48] L.-W. Wang and A. Zunger, Phys. Rev. B 51, 17398 (1995).
- [49] H. Haug and A.-P. Jauho, *Quantum Kinetics in Transport and Optics of Semiconductors* (Springer, Berlin, 2008).
- [50] S. Sanvito, C. J. Lambert, J. H. Jefferson, and A. M. Bratkovsky, Phys. Rev. B 59, 11936 (1999).
- [51] P. A. Khomyakov, G. Brocks, V. Karpan, M. Zwierzycki, and P. J. Kelly, Phys. Rev. B 72, 035450 (2005).
- [52] H. H. B. Sørensen, P. C. Hansen, D. E. Petersen, S. Skelboe, and K. Stokbro, Phys. Rev. B 77, 155301 (2008).
- [53] M. P. L. Sancho, J. M. L. Sancho, J. M. L. Sancho, and J. Rubio, J. Phys. F 15, 851 (1985).
- [54] D. E. Petersen, H. H. B. Sørensen, P. C. Hansen, S. Skelboe, and K. Stokbro, J. Comput. Phys. 227, 3174 (2008).
- [55] P. R. Amestoy, I. S. Duff, J.-Y. L'Excellent, and J. Koster, SIAM J. Matrix Anal. Appl. 23, 15 (2001).
- [56] L. Lin, M. Chen, C. Yang, and L. He, J. Phys.: Condens. Matter 25, 295501 (2013).
- [57] J. P. Perdew, K. Burke, and M. Ernzerhof, Phys. Rev. Lett. 77, 3865 (1996).
- [58] H. J. Monkhorst and J. D. Pack, Phys. Rev. B 13, 5188 (1976).
- [59] The electrostatic difference potential is defined as $\delta V_{\rm E} = V_{\rm E}^{\rm scf} V_{\rm E}^{\rm sp}$, where $V_{\rm E}^{\rm scf}$ is the electrostatic potential of the self-consistent valence charge density and $V_{\rm E}^{\rm sp}$ the electrostatic potential from a superposition of atomic valence densities, $n^{\rm sp}$. Similarly, we define the Hartree difference potential as $\delta V_{\rm H} = V_{\rm H}^{\rm scf} V_{\rm H}^{\rm sp} = -e\delta V_{\rm E}$, where *e* is the elementary charge (e > 0), and the electron difference density as $\delta n = n^{\rm scf} n^{\rm sp}$.
- [60] Electron Work Function of the Crystalline Elements, in CRC Handbook of Chemistry and Physics, 97th ed. (CRC Press/Taylor & Francis, Boca Raton, FL, 2017).
- [61] G. Giovannetti, P. A. Khomyakov, G. Brocks, V. M. Karpan, J. Van Den Brink, and P. J. Kelly, Phys. Rev. Lett. 101, 026803 (2008).
- [62] S. Trasatti, J. Electroanal. Chem. 33, 351 (1971).
- [63] J. Neugebauer and M. Scheffler, Phys. Rev. B 46, 16067 (1992).
- [64] A. Baldereschi, S. Baroni, and R. Resta, Phys. Rev. Lett. 61, 734 (1988).
- [65] S. García-Gil, A. García, N. Lorente, and P. Ordejón, Phys. Rev. B 79, 075441 (2009).
- [66] The details on the doping methods implemented in the ATK code can be found in the ATK technical notes at http://docs.quantumwise.com/technicalnotes.html.
- [67] J. Tersoff, Phys. Rev. B 30, 4874 (1984).
- [68] K. Berland, T. L. Einstein, and P. Hyldgaard, Phys. Rev. B 85, 035427 (2012).
- [69] W. Shockley, Phys. Rev. 56, 317 (1939).

- [70] H. Zhang, C.-X. Liu, X.-L. Qi, X. Dai, Z. Fang, and S.-C. Zhang, Nat. Phys. 5, 438 (2009).
- [71] Y. Xia, D. Qian, D. Hsieh, L. Wray, A. Pal, H. Lin, A. Bansil, D. Grauer, Y. S. Hor, R. J. Cava, and M. Z. Hasan, Nat. Phys. 5, 398 (2009).
- [72] L. Limot, T. Maroutian, P. Johansson, and R. Berndt, Phys. Rev. Lett. 91, 196801 (2003).
- [73] J. Kröger, L. Limot, H. Jensen, R. Berndt, and P. Johansson, Phys. Rev. B 70, 033401 (2004).
- [74] W. Zhang, R. Yu, H.-J. Zhang, X. Dai, and Z. Fang, New J. Phys. 12, 065013 (2010).
- [75] S. D. Kevan and R. H. Gaylord, Phys. Rev. B 36, 5809 (1987).
- [76] M. Garnica, M. Schwarz, J. Ducke, Y. He, F. Bischoff, J. V. Barth, W. Auwärter, and D. Stradi, Phys. Rev. B 94, 155431 (2016).
- [77] O. V. Yazyev, E. Kioupakis, J. E. Moore, and S. G. Louie, Phys. Rev. B 85, 161101 (2012).
- [78] J. Neugebauer and M. Scheffler, Surf. Sci. 287, 572 (1993).
- [79] S. A. Wasileski, M. T. M. Koper, and M. J. Weaver, J. Chem. Phys. 115, 8193 (2001).
- [80] M. Otani and O. Sugino, Phys. Rev. B 73, 115407 (2006).
- [81] N. Bonnet, I. Dabo, and N. Marzari, Electrochim. Acta 121, 210 (2014).
- [82] R. Gomer, Surf. Sci. 299, 129 (1994).
- [83] C. G. Sánchez, A. Y. Lozovoi, and A. Alavi, Mol. Phys. 102, 1045 (2004).
- [84] A. Tkatchenko, N. Batina, A. Cedillo, and M. Galván, Surf. Sci. 581, 58 (2005).
- [85] B. C. Schardt, S.-L. Yau, and F. Rinaldi, Science 243, 1050 (1989).

- [86] B. Zhang, D. Wang, Y. Hou, S. Yang, X. H. Yang, J. H. Zhong, J. Liu, H. F. Wang, P. Hu, H. J. Zhao, and H. G. Yang, Sci. Rep. 3, 1836 (2013).
- [87] J. Wellendorff, T. L. Silbaugh, D. Garcia-Pintos, J. K. Nørskov, T. Bligaard, F. Studt, and C. T. Campbell, Surf. Sci. 640, 36 (2015).
- [88] S. F. Boys and F. Bernardi, Mol. Phys. 19, 553 (1970).
- [89] A. Garcia-Lekue, D. Sánchez-Portal, A. Arnau, and L. W. Wang, Phys. Rev. B 88, 155441 (2013).
- [90] L. Kleinman and D. M. Bylander, Phys. Rev. Lett. 48, 1425 (1982).
- [91] V. Blum, R. Gehrke, F. Hanke, P. Havu, V. Havu, X. Ren, K. Reuter, and M. Scheffler, Comput. Phys. Commun. 180, 2175 (2009).
- [92] S. G. Louie, S. Froyen, and M. L. Cohen, Phys. Rev. B 26, 1738 (1982).
- [93] G. Theurich and N. A. Hill, Phys. Rev. B 64, 073106 (2001).
- [94] K. Lejaeghere, V. Van Speybroeck, G. Van Oost, and S. Cottenier, Crit. Rev. Solid State Mater. Sci. **39**, 1 (2014).
- [95] Comparing Solid State DFT Codes, Basis Sets and Potentials, https://molmod.ugent.be/deltacodesdft.
- [96] K. F. Garrity, J. W. Bennett, K. M. Rabe, and D. Vanderbilt, Comput. Mater. Sci. 81, 446 (2014).
- [97] L. Schimka, J. Harl, and G. Kresse, J. Chem. Phys. 134, 024116 (2011).
- [98] O. Madelung, *Semiconductors: Data Handbook*, 3rd ed. (Springer, Berlin, 2004).
- [99] F. Schaffler, in *Properties of Advanced Semiconductor Materials: GaN, AlN, InN, BN, SiC, SiGe*, edited by M. E. Levinshtein, S. L. Rumyantsev, and M. S. Shur (Wiley, New York, 2001) Chap. 6, pp. 149–188.
- [100] A. V. Krukau, O. A. Vydrov, A. F. Izmaylov, and G. E. Scuseria, J. Chem. Phys. **125**, 224106 (2006).